Author's response to reviews

Title: A study to investigate the effectiveness of SimMan as an adjunct in teaching preclinical skills to medical students

Authors:

Meenakshi Swamy (meenakshi.swamy@durham.ac.uk)
Marina Sawdon (marina.sawdon@durham.ac.uk)
Andrew Chaytor (andrew.chaytor@durham.ac.uk)
David Cox (d.r.cox@durham.ac.uk)
Judith Barbaro-Brown (j.a.barbaro-brown@durham.ac.uk)
John McLachlan (j.c.mclachlan@durham.ac.uk)

Version: 2
Date: 6 March 2014

Author's response to reviews:

I have included the changes requested in the covering letter